Author’s response to reviews

Title: Disciplinary boundaries and integrating care: using Q-methodology to understand trainee views on being a good doctor

Authors:

Esther Muddiman (muddimanek@cardiff.ac.uk)
Alison Bullock (BullockAD@Cardiff.ac.uk)
Jennifer Hampton (hamptonjm1@cardiff.ac.uk)
Lynne Allery (Allery@cardiff.ac.uk)
Janet MacDonald (MacdonaldJ@cardiff.ac.uk)
Katie Webb (webbk11@cardiff.ac.uk)
Lesley Pugsley (pugsleyla@cardiff.ac.uk)

Version: 1 Date: 22 Nov 2017

Author’s response to reviews:

Dear Kerry

Many thanks for your email and sincere apologies that I have only been able to attend to these issues now - I found your correspondence in my junk mail folder.

I hope that the revisions made are suitable, please do let me know if there is anything else I can do.

Best wishes

Esther